Author's response to reviews

Title: The identification and epidemiology of methicillin-resistant Staphylococcus aureus and Clostridium difficile in patient rooms and the ward environment

Authors:

Meredith C Faires (mfaires@uoguelph.ca)
David L Pear (dpearl@uoguelph.ca)
Olaf Berke (oberke@uoguelph.ca)
Richard J Reid-Smith (rrreidsmi@ovc.uoguelph.ca)
J Scott Weese (jsweese@ovc.uoguelph.ca)

Version: 2 Date: 23 April 2013

Author's response to reviews: see over
Dear Editor,

Please find enclosed the revised manuscript “The identification and epidemiology of methicillin-resistant *Staphylococcus aureus* and *Clostridium difficile* in patient rooms and the ward environment” by Faires et al.

As per your request, information pertaining to informed consent from patients participating in our study has been included in the Methods section under the heading “2.1 Study site”.

If you have any other questions or concerns regarding the above manuscript, please do not hesitate to contact me.

Yours sincerely,

Dr. Meredith Faires